ANSWERING REVIEWERS

This very well-written manuscript giving a historical retrospective analysis, illustrates very clearly the usefulness and effect of the Pouwels classification on the development of hip surgery.

_We thank the reviewers for their outstanding reviews. We appreciate the constructive feedback._

The “Author Contributions” section is missing. Please provide the author contributions

_The author contributions have been recorded as advised._

PMID and DOI numbers are missing in the reference list. Please provide the PubMed numbers and DOI citation numbers to the reference list and list all authors of the references. Please revise throughout.

_The PMID and DOI numbers have been included in the reference list. All authors for each reference have been listed. We utilized the reference auto-analyser._